Author's response to reviews

Title: Renal cell carcinoma metastasis to the ciliary body responds to proton beam radiotherapy

Authors:

Tarek Alasil (tarekalasil@hotmail.com)
Bahram Khazai (bahram_khazaee@yahoo.com)
Lilia Loredo (lloredo@dominion.llumc.edu)
Michael E Rauser (mrauser@llu.edu)

Version: 2 Date: 7 April 2011

Author's response to reviews: see over
Dear Editors at JMCR,
I would like to thank you for considering our manuscript named “Renal Cell Carcinoma metastasis to the ciliary body responds to proton beam radiotherapy” for publication at the JMCR. I appreciate the valuable comments and suggestions from the reviewers.

Attached, please find the revised version of the manuscript. We addressed the reviewers’ valuable comments. You will be able to track those changes in red throughout the manuscript.

Reviewer 1:
PROVIDE DETAILS OF THE PROTON THERAPY: DOSE, FRACTIONS, ETC.
The proton treatment details were added. Please see page 5; lines 11-14.

Reviewer 2:
1. Unexpected or unusual presentations of a disease.
The word was changed to Unexpected. Please see page 2; line 3.

2. Please express more details about the pathologic and histological features (e.g. Fuhrman grade of metastatic neoplasm)
Please see page 14; lines 3-5.

3. Provide more information about the collaboration with urologists. How was the follow-up schedule during the 10 years and later? Do they respect the guidelines?
Please see page 4; lines 6-8.
Please see page 7; lines 6-9.
Also please see references 15-20.